Author's response to reviews

Title: A neonatal presentation of factor V deficiency: A case report

Authors:

Amol Chingale (dramolchingale@hotmail.com)
Michael Eisenhut (michael_eisenhut@yahoo.com)
Anjali Gadiraju (rajuanju1@yahoo.co.in)
Ri Liesner (liesnr@gosh.nhs.uk)

Version: 5 Date: 20 February 2007

Author's response to reviews:

Dear Editor,
Formatting changes have been made as recommended in the production teams e-mail dated 20/02/2007. Some typographical errors in the manuscript text have also been corrected.
On behalf of all authors
Regards
Michael Eisenhut